Author's response to reviews

Title: Smoking, mental illness and socioeconomic disadvantage: Analysis of the Australian National Survey of Mental Health and Wellbeing

Authors:

David Lawrence (dlawrence@ichr.uwa.edu.au)
Jennifer Hafekost (jhafekost@ichr.uwa.edu.au)
Philip Hull (philh@nswcc.org.au)
Francis Mitrou (francism@ichr.uwa.edu.au)
Steve Zubrick (steve@ichr.uwa.edu.au)

Version: 2 Date: 16 November 2012

Author's response to reviews:

Thankyou for your question as to whether the data used in our paper is openly available. We have added the following sentence to the manuscript:

"As the study consisted of analysis of publicly available confidentialised files, no institutional ethics approval was required."